Author's response to reviews

Title: The Vaquez’ disease revealed by bladder tumor in erectile disorder patient, Case report

Authors:

Ahmed Amine Bouchikhi (drbouchikhi@gmail.com)
MF Tazi (mft@lycos.com)
Driess Amiroune (d.amiroune@gmail.com)
JE Elammari (jea@hotmail.com)
A Khallouk (abKh@live.fr)
S Mellass (smel@lycos.com)
MJ Elfassi (elfa@live.fr)
MH Farih (far@hotmail.fr)

Version: 2 Date: 20 September 2012

Author's response to reviews: see over
Dear Editor of JMCR,

We are submitting our paper entitled “The Vaquez' disease revealed by bladder tumor in erectile disorder patient, Case report”


We are estimating that this paper would contribute to clarifying attitudes in the elucidated disease. Evermore, we do think that this case has raised issues concerning the confirmation of therapeutical approach to be conducted while dealing with the association of Vaquez' disease, bladder tumor and erectile disorder. Indeed it is the first case in the literature associating the mentioned pathological components. In addition, the particular clinical features are of great interest for clinicians dealing with any pathological features discussed in the paper. In fact, this paper has multiple components that are of interests for urologists, internal medicine, anesthesia, sexologists, oncologists, general practitioners.

We do hope that our manuscript will find the way to publication in your journal and kind regards,

Dr. Ahmed Amine Bouchikhi